Author's response to reviews

Title: Diffuse bronchiolitis pattern on CT scan as a presentation of pulmonary tumor thrombotic microangiopathy: a case report

Authors:

Maria Fernanda Arruda MFA (mf_arruda@yahoo.com.br)
André Brelinger AB (andre_brelinger@yahoo.com.br)
Paula Nicole Barbosa PNB (nicolebarbosa@terra.com.br)
Rubens Chojniak RC (chojniak@uol.com.br)
Jefferson Luiz Gross JLG (jefluizgross@yahoo.com.br)
Marcos Duarte Guimaraes MDG (marcosduarte500@gmail.com)

Version: 4 Date: 18 October 2011

Author's response to reviews:

The authors have accepted all the corrections suggested by the reviewers and the editor making the necessary adjustments in the manuscript. All adjustments were made in the manuscript text.